Author's response to reviews

Title: The Effect of Verbalization Strategy on Wisconsin Card Sorting Test Performance in Schizophrenic Patients Receiving Classical or Atypical Antipsychotics

Authors:

Alessandro Rossi (alessandro.rossi@cc.univaq.it)
Enrico Daneluzzo (cim.alba@asl18.sanitacn.it)
Annarita Tomassini (annaritatomassini@yahoo.it)
Francesca Struglia (francescastruglia@hotmail.com)
Roberto Cavallaro (cavallaro.roberto@hsr.it)
Enrico Smeraldi (smeraldi@hsr.it)
Paolo Stratta (psystr@tin.it)

Version: Date: 26 November 2005

Author's response to reviews:

L'Aquila, November 26, 2005

BioMed Central

Re: Manuscript "THE EFFECT OF VERBALIZATION STRATEGY ON WISCONSIN CARD SORTING TEST PERFORMANCE IN SCHIZOPHRENIC PATIENTS RECEIVING CLASSICAL OR ATYPICAL ANTIPSYCHOTICS"

Dear Editor,

I am writing to submit the manuscript after having made the suggested modifications.

* We give now more information about review board: Ethic Committee A.U.S.L. 4 L'Aquila (pg. 5).

* Authors' contribution paragraph has been added (pg. 9).

Reviewer Janusz K Rybakowski

The suggested changes in the tables have been made.

I hope that the paper could be accepted for publication.

Yours faithfully,

Prof. Alessandro Rossi